Author's response to reviews

Title: Myasthenia gravis and pregnancy: clinical implications and neonatal outcome

Authors:

Jose Francisco Tellez-Zenteno (jftellez@quetzal.innsz.mx)
Lizbeth Hernandez-Ronquillo (liztellez33@hotmail.com)
Vicente Salinas (vicente_salinas@latinmail.com)
Bruno Estanol (bestanol@hotmail.com)
da Silva Orlando (odasilva@julian.uwo.ca)

Version: 9 Date: 15 September 2004

Author's response to reviews: see over
Author list – We changed the order of the name of the authors according submission system.
Abstract - We changed the abstract according your recommendation and we changed the abstract in the submission page
Keywords section – We deleted the key words in the article
Section headings- We removed the unnecessary capitalization
References – We used the square brackets.
Competing interests – There are no competing interest to declare
Authors' contributions - We have added an Authors' Contributions section
Acknowledgements- We include the acknowledgments in the section Authors' contributions